Author’s response to reviews

Title: Performance Feedback: An exploratory study to examine the acceptability and impact for interdisciplinary primary care teams

Authors:
Sharon Johnston (sjohnston@bruyere.org)
Green Michael (michael.green@dfm.queensu.ca)
Patricia Thille (pthille@uottawa.ca)
Colleen Savage (savagec@queensu.ca)
Lynn Roberts (roberts@queensu.ca)
Grant Russell (Grant.Russell@med.monash.edu.au)
William Hogg (whogg@uottawa.ca)

Version: 3 Date: 24 January 2011

Author’s response to reviews: see over
Editor, BMC Family Practice

January 24th, 2011
Dear Editor,

**Re: Manuscript Re-Submission**

Thank you for the opportunity to re-submit our manuscript titled, *Performance Feedback: An exploratory study to examine the acceptability and impact for interdisciplinary primary care teams*, for consideration by BMC Family Practice. We appreciate the second opportunity to address the constructive feedback from the reviewers. This paper presents a mixed methods study that was designed to explore the acceptability and impact of feedback of team performance data to primary care interdisciplinary teams. It demonstrates that performance feedback to interdisciplinary teams may build their capacity for quality improvement.

The following authors have contributed to the writing of this manuscript and approved the final version: Drs. Johnston, Green, Thille, Savage, Roberts, Russell and Hogg.

The most recent requested revisions have been made and the changes have been tracked in the revised manuscript. We have addressed the reviewer’s and associate editor’s comments as follows:

- The results section has been shortened, to be more focused in order to maintain reader interest.
- Tables 6 and 7 have been updated to include the numbers of valid responses by question, in accordance with the editor’s specifications.

I confirm that all authors have seen and agree with the contents of this manuscript and confirm that the work has not been submitted or published elsewhere in whole or in part.

This manuscript is submitted in accordance with the regulations as defined by the BMC Family Practice editorial policies.

Please do not hesitate to contact me if any further information is required.

Sincerely,

Dr. Sharon Johnston – Corresponding author
Assistant Professor
Department of Family Medicine
University of Ottawa
CT Lamont Primary Health Care Research Centre
Élisabeth Bruyère Research Institute
43 Bruyère Street
Ottawa, On, K1N 5C8
Tel: (613) 562-0050 ext.1599